Author’s response to reviews

Title: In vivo dual targeting of the oncogenic Ether-a-go-go-1 potassium channel by calcitriol and astemizole results in enhanced antineoplastic effects in breast tumors

Authors:

- Janice García-Quiroz (janicegarciaq@gmail.com)
- Rocío García-Becerra (rocio.garciab@quetzal.innsz.mx)
- Nancy Santos-Martínez (santos_2105@hotmail.com)
- David Barrera (barrera1912@gmail.com)
- David Ordaz-Rosado (davo21_05@hotmail.com)
- Euclides Avila (euclides.avilac@incmnsz.mx)
- Ali Halhali (ali.halhalib@incmnsz.mx)
- Octavio Villanueva (octavio610216@aol.com)
- María J Ibarra-Sánchez (mary.ibarra@mail.mcgill.ca)
- José Esparza-López (jesparza@biomedicas.unam.mx)
- Armando Gamboa-Domínguez (agamboad@gmail.com)
- Javier Camacho (fcamacho@cinvestav.mx)
- Fernando Larrea (fernando.larreaq@quetzal.innsz.mx)
- Lorenza Díaz (lorenzadiaz@gmail.com)

Version: 4
Date: 17 September 2014

Author’s response to reviews: see over
Author’s response to Editor

Title: In vivo dual targeting of the oncogene KCNH1 by calcitriol and astemizole results in enhanced antineoplastic effects in breast tumors

Authors:

JGQ: janicegarciaq@gmail.com
RGB: rocio.garciab@quetzal.innsz.mx
NSM: santos_2105@hotmail.com
DBH: barrera1912@gmail.com
DOR: davo21_05@hotmail.com
EA: euclides.avilac@incmnsz.mx
AH: ali.halhalib@incmnsz.mx
OV: octavio610216@aol.com
MJIS: mary.ibarra@mail.mcgill.ca
JEL: jesparza@biomedicas.unam.mx
JC: fcamacho@cinvestav.mx
FL: fernando.larreag@quetzal.innsz.mx
LD: lorenzadiaz@gmail.com

Version: 3 Date: 17 September 2014
Editorial requests:

------------------

1) Please could you additionally declare in your competing interest section that Hoffmann-La Roche Ltd donated calcitriol.

2) Please remove your response to reviewers from additional files.

------------------:

1) We have added in the competing interest section the following sentence: “Hoffmann-La Roche Ltd kindly donated calcitriol for this study”.

In addition, we added the following paragraph in the acknowledgments section: “We acknowledge with thanks to Sofia Campuzano for photographic assistance.”

2) We removed the response to reviewers from additional files.